Title: Diabetes in Rural Towns: Protocol for a cluster randomised controlled trial of the effectiveness of continuing education and feedback for health care providers in altering diabetes outcomes at a population level.

Authors:

Christine L Paul (chris.paul@newcastle.edu.au)
Leon Piterman (leon.piterman@monash.edu)
Jonathan Shaw (Jonathan.Shaw@bakeridi.edu.au)
Catherine Kirby (Catherine.Kirby@monash.edu)
Robert W Sanson-Fisher (Rob.Sanson-Fisher@newcastle.edu.au)
Mariko L Carey (Mariko.Carey@newcastle.edu.au)
Jennifer Robinson (Jennifer.Robinson@newcastle.edu.au)
Patrick McElduff (Patrick.Mcelduff@newcastle.edu.au)
Isaraporn Thepwongsa (isaraporn.thepwongsa@monash.edu)

Version: 4 Date: 4 March 2013

Author's response to reviews:

There is just one change which is incorrect – an ‘e’ was missing from ‘effectiveness’ in the title on the first page.